Author's response to reviews

Title: Burden of severe RSV disease among immunocompromised children and adults: a 10 year

retrospective study.

Authors:

Olga Chatzis (olga.chatzis@uclouvain.be)

Stephanie Darbre (stephanie.darbre@chuv.ch)

Jérôme Pasquier (jerome.pasquier@chuv.ch)

Pascal Meylan (pascal.meylan@chuv.ch)

Oriol Manuel (oriol.manuel@chuv.ch)

John-David Aubert (john-david.aubert@chuv.ch)

Maja Beck-Popovic (maja.beck-popovic@chuv.ch)

Stavroula Masouridi-Levrat (Stavroula.Masouridi@hcuge.ch)

Marc Ansari (marc.ansari@hcuge.ch)

Laurent Kaiser (laurent.kaiser@hcuge.ch)

Klara Posfay-Barbe (klara.posfay-barbe@hcuge.ch)

Sandra Asner (sandra.asner@chuv.ch)

Version: 3 Date: 06 Feb 2018

Author's response to reviews:

Replies to the reviewers' comments

We thank the editor for his/her comments and for giving us the opportunity to address all minor

edits. Please note that the page and line numbers in our responses to reviewers' comments refer

to our numbering in the revised manuscript.

Editor Comments

• Please include the email addresses for all authors on the title page. The corresponding author should still be indicated. Please also ensure all email addresses and names of authors provided in the manuscript and the submission system match.

We have provide all email addresses on the title page 1/15.

• Please ensure that all tables/figures/supplementary files are cited within the main text. Currently, Table 2 is missing.

We apologize for the inconvenience. As Table 2 is cited l.9, p 8, we have added back Table 2 to the "Table" document.

• The individual contributions of ALL authors to the manuscript should be specified in the Authors' Contributions section. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

We apologize for the inconvenience. We have specified the contribution of all authors in the Author's Contribution section, p.12

• As only summarized data is presented in this manuscript, therefore we ask that you amend the data availability statement. This statement refers to the raw data used in your study and presenting tables and is not sufficient to state that all data is contained within the manuscript and additional files. Please either include all raw data obtained in this study or clearly state that the data is available upon request and who should be contacted if someone wants to request the data.

We have clearly stated that the data is available upon request and that Dr Sandra A. Asner should be contacted if someone wants to request the data.

• We note that you have not included an acknowledgements section. If you have no acknowledgements please put 'Not Applicable' in this section.

We have completed the acknowledgement section by stating "Not applicable"